University of Waterloo STAT 332 – W. H. Cherry

#2.131a

Figure 2.17. UNSTRATIFIED POPULATIONS: Ratio Estimating for an Average or a Total Section headings to provide a content overview

One-Stage EPSWOR of Individual Elements

1. Background 2. Ratio Estimating: Point Estimates and Standard Deviations 3. Ratio Estimating: Estimating Bias

Example 2.17.1 4. Efficiency

Example 2.17.2 Example 2.17.3 5. Appendix 1: How Does Ratio Estimating Increase Precision? 6. Appendix 2: The Sample Ratio as a Linear Model Slope Parameter 7. Reference Three tables Two diagrams Thirty-two equations

2025-08-20